
Tuesday: Jim Morgan•
Wednesday: Tony Leveling•
Thursday: Jim Morgan•
Friday: Keith Gollwitzer•
Saturday:  Tony Leveling•
Sunday: Dave Vander Muelen•

On-call

The JASMIN air gap samples with tubes and pull ropes have also been installed.○

JASMIN shield blocks to more time to install than our old shielding blocks.  All have been 
installed at this time except for the final keystone block. It does not fit. The problem is that 
the overhang of the first block at the south end of the vault does extend far enough due to 
the vault wall. The clearance available at the north end is somewhere between 1 and 2 
inches short. We are installing a stack of three 19" X 66" X 148" blocks at the end of the 
vault shield. We need to add some additional hand stack blocks to form an air dam around 
this last set of blocks. A total of 20 handstack blocks are required and are being gathered 
from SBSB at this time. The RSO has examined the alternative shield plan and agrees that it 
is a conservative measure.Pasted from <http://www-bd.fnal.gov/cgi-mach/machlog.pl?
nb=pts&action=view&page=last&frame=2&scroll=true&load=> 

○

JASMIN shielding blockwork•

Checked the beam dump reservoir tank level today AFTER a water sample was collected. The level 
was down about 3/8" to 7/16". Pasted from <http://www-bd.fnal.gov/cgi-mach/machlog.pl?
nb=pbar10&action=view&page=276&frame=2&anchor=&hilite=&load=> 

•

We also flushed the collimator return line strainer during the target change today. Flow increased 
from about 4 to 4.25 gpm. Pasted from <http://www-bd.fnal.gov/cgi-mach/machlog.pl?
nb=pts&action=view&page=last&frame=2&scroll=true&load=> 

•

PreVault access take more pictures•

Found target damage not limited to small area ○

Target 16 was installed on February 9, 2010. There were 5.26E19 protons on target over this 
period. Target 16 was the second best target so far; it took about 90% of the total protons 
that Target 14 had, but with a significantly reduced spot size over its entire service life. 
Pasted from <http://www-bd.fnal.gov/cgi-mach/machlog.pl?
nb=pts&action=view&page=last&frame=2&scroll=true&load=> 

○

Some evidence of sputtering on Li Lens and Target SEM○

Took pictures of target •

Target change went smoothly.•
Update on window•

Downtime

<stacking rate>=26.4mA/hr○

<production>=23.3 pbar/E6 p○

<POT>= 8.34 Tp○

Stacking Numbers•

Stacking

Unstacked 193e10 in 24 transfers over 8 sets.•

Transfers
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Unstacked 193e10 in 24 transfers over 8 sets.•
Overall Accumulator to Recycler transfer efficiency 93.6%.•
Transfer 21364 (01:50am) had only  a 75%.  Ops had some problems with the A10 BPMs and didn't 
get a good tune-up.   A reboot of the BPM house helped.

•

Removing 21364, we had an overall efficiency of 96.8%•
• Reported to Vaia that transfers 21361, 21362 and 21363 all have the injection MI DCCT number at 

about 50% of it's expected value.  This makes the Accumulator to MI number come out wrong.

Move shot setup TLGs to 2.6 second cycle time•
Test 2.4 and 2.2 second cycle times•

Requests

Pbars stacked: 201.43 E10○

Time stacking: 08.28 Hr○

Average stacking rate: 24.32 E10/Hr○

Stacking•

Number of pulses while in stacking mode: 10881○

Number of pulses with beam: 10234○

Fraction of up pulses was: 94.05%○

Uptime•

Corrected time stacking: 07.79 Hr○

Possible average stacking rate: 25.86 E10/Hr○

Could have stacked: 214.16 E10/Hr○

The uptime's effect on the stacking numbers•

Pbars sent to the Recycler: 193.40 E10○

Number of transfers : 25○

Number of transfer sets: 8○

Average Number of transfer per set: 3.12○

Time taken to shoot including reverse proton tuneup: 00.09 Hr○

Transfer efficiency: 93.92%○

Recycler Transfers•

Average POT : 8.34 E12○

Average production: 23.59 pbars/E6 protons○

Other Info•

* Red indicates a problem during data retrieval.  See the message window for details.•

The Numbers

The Plots
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The Plots
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